Author’s response to reviews

Title: Field-testing of the Rapid Assessment of Disability Questionnaire

Authors:

Manjula Marella (marella.m@unimelb.edu.au)
Lucy Busija (l.busija@deakin.edu.au)
Amirul Islam (amirulnarail@gmail.com)
Alexandra Devine (devinea@unimelb.edu.au)
Kathy Fotis (kfotis@unimelb.edu.au)
Sally Baker (sm.baker@unimelb.edu.au)
Beth Sprunt (b.sprunt@unimelb.edu.au)
Tanya Edmonds (tedmonds@unimelb.edu.au)
Huq L Nafisa (lira@icddrb.org)
Anaseini Cama (anaseini2001@yahoo.com)
Jill Keeffe (jill.keeffe@unimelb.edu.au)

Version: 3 Date: 10 January 2014

Author's response to reviews:

Rita Aguirre,
Journal Editorial Office
BMC Public Health

Dear Rita,

Please see our point-by-point response to the suggested changes.

1. Title page: As requested email addresses of all authors have been included in the Title page along with the authors’ names, institutions and countries and the full postal address of the corresponding author.

2. Acknowledgements: As indicated an acknowledgements section has been included between the Authors’ contributions section and Reference list.

In addition, the Authors’ contributions section has been revised to cut down the word count.

This paper is not currently being considered elsewhere, is not in press in any format, and has not been previously published. No author has any conflict of interest related to this paper.

Thank you for your kind attention and we look forward to a favourable reply.

Kind regards,

Dr Manjula Marella
Senior Research Officer,
Nossal Institute for Global Health, The University of Melbourne, Level 4, Alan Gilbert Building, 161 Barry Street, Carlton, VIC 3010
Tel: + 61 3 9035 3067, Fax: + 61 3 9347 6872
Email: marella.m@unimelb.edu.au